# communications medicine

### ARTICLE

https://doi.org/10.1038/s43856-022-00145-0

OPEN

# Modeling the impact of COVID-19 on future tuberculosis burden

Mario Tovar <sup>[</sup><sup>0</sup>,<sup>2</sup><sup>⊠</sup>, Alberto Aleta <sup>[</sup><sup>9</sup>, Joaquín Sanz<sup>1,2</sup> & Yamir Moreno <sup>[</sup><sup>1,2,3</sup>

#### Abstract

**Background** The ongoing COVID-19 pandemic has greatly disrupted our everyday life, forcing the adoption of non-pharmaceutical interventions in many countries and putting public health services and healthcare systems worldwide under stress. These circumstances are leading to unintended effects such as the increase in the burden of other diseases.

**Methods** Here, using a data-driven epidemiological model for tuberculosis (TB) spreading, we describe the expected rise in TB incidence and mortality if COVID-associated changes in TB notification are sustained and attributable entirely to disrupted diagnosis and treatment adherence.

**Results** Our calculations show that the reduction in diagnosis of new TB cases due to the COVID-19 pandemic could result in 228k (CI 187-276) excess deaths in India, 111k (CI 93-134) in Indonesia, 27k (CI 21-33) in Pakistan, and 12k (CI 9-18) in Kenya.

**Conclusions** We show that it is possible to reverse these excess deaths by increasing the pre-covid diagnosis capabilities from 15 to 50% for 2 to 4 years. This would prevent almost all TB-related excess mortality that could be caused by the COVID-19 pandemic if no additional preventative measures are introduced. Our work therefore provides guidelines for mitigating the impact of COVID-19 on tuberculosis epidemic in the years to come.

#### Plain language summary

The COVID-19 pandemic has disrupted everyday life and put public health services and healthcare systems worldwide under stress. This has compromised the ability to control other diseases such as Malaria, Cancer and Tuberculosis. In this work we predict the rise in Tuberculosis occurrence and mortality when healthcare systems are impacted and diagnosis capabilities blocked in 4 countries where TB is prevalent. Our calculations show that an increase in new TB cases due to the COVID-19 pandemic could result in almost 400,000 additional deaths from TB in India, Indonesia, Pakistan and Kenva. We also show that increased diagnosis capabilities after the pandemic could reduce the additional deaths from TB resulting from the COVID-19 pandemic impact.

<sup>1</sup> Institute for Biocomputation and Physics of Complex Systems (BIFI), University of Zaragoza, 50009 Zaragoza, Spain. <sup>2</sup> Department of Theoretical Physics, University of Zaragoza, 50009 Zaragoza, Spain. <sup>3</sup> ISI Foundation, Via Chisola 5, 10126 Torino, Italy. <sup>ISI</sup> email: mariotovarzgz@gmail.com



uberculosis (TB) is an infectious disease caused by the bacterium Mycobacterium tuberculosis (M.tb.) that usually affects the lungs. It is a preventable but complex disease with a high global burden that requires early detection and long treatments. Despite the global effort to eradicate TB and recent decreases in its burden due to the implementation of strategies aimed at optimizing diagnosis and treatment<sup>1,2</sup>, it remains one of the greatest threats to public health worldwide, being the deadliest single-agent persistent infectious disease nowadays. According to the 2021 Global TB Report by the World Health Organization (WHO)<sup>3</sup>, ten million people developed TB and nearly 1.5 million people died because of TB infection in 2020, and for the first time in a decade, there is an increase in TB-caused deaths. In the last decades, the WHO has deployed a series of global strategies that have since been the backbone of the global fight against TB. In 1995, the Directed Observed Treatment Strategy was introduced, which significantly strengthened the capacity of national programs to diagnose and treat TB cases. Later, the Stop TB Strategy, announced in 2006, was the first of such plans at setting a TB elimination horizon, defined as a reduction of incidence levels under one case per million and year by 2050. A redefinition of the eradication goal took place in 2014 when the previous objective was moved forward to 2035 within the End TB Strategy.

If the elimination target set by the End TB strategy was already an ambitious goal<sup>4</sup>, the emergence of the COVID-19 pandemic caused by the new coronavirus SARS-CoV-2 sheds significant concerns on whether these goals are still reachable. During the acute stages of the COVID-19 pandemic, economic and human resources were redirected to control and mitigate the emergency caused by the pandemic, which led to a great reduction in the diagnosis of new cases of other diseases, as already documented for cancer, or malaria<sup>5,6</sup>. Interventions such as long lockdowns and mobility restrictions have exacerbated shortages in resources otherwise destined for the care of patients suffering these, and other pathologies. Moreover, COVID-19 has greatly affected healthcare workers<sup>7–9</sup>, thus creating additional pressure to healthcare systems.

TB diagnosis and patient care are no exceptions, as reported in previous literature<sup>3,10,11</sup>. As a primary and immediate effect of COVID-19 spreading onto TB transmission dynamics, a reduction in the case notification ratio has been observed during and immediately after lockdowns and periods of high COVID-19 incidence and saturation of healthcare facilities<sup>3</sup>. We hypothesize that this disruption alone will lead to a surge in TB burden in the next years, even before the more complex, and less predictable effects of the COVID-19 pandemic on TB management and transmission dynamics can be properly characterized. For example, drastic drops in laboratory capacity needed to support TB diagnosis are expected along with interruptions in the supply of drugs, which could result in shortages of medications and could delay the start of treatments until the supply chain is reestablished<sup>12-14</sup>. Moreover, as suggested by Cilloni et al.<sup>15</sup>, even temporary stoppages might cause long-term increases in TB incidence and mortality, and a peak in TB burden is to be observed as a consequence of the healthcare system disruption.

In this work, we assess the impact of COVID-19 on the expected TB burden until the year 2035, which marks the target horizon of the End TB Strategy. Specifically, we incorporate the observed drop in TB diagnosis and treatment compliance rates caused by COVID-19 into a mathematical model that produces long-term forecasts of TB burden<sup>16</sup>. This allows us to: (1) quantify the effect of the COVID-19 stoppage in relation to a baseline scenario in which no pandemic happened, and (2) compute the effect that a rapid response to the uprising TB

burden in the following years, in the form of a compensatory intervention aiming at boosting TB diagnosis rates as soon as the COVID-19 pandemic ends, has over long-term TB goals. Our results show that an effort focused on increasing TB diagnosis capabilities once the pandemic is over could revert the effect of the pandemic in the long term.

#### Methods

**Model calibration and diagnosis rate.** In this study, we have capitalized on the detailed M.tb. transmission model developed by Arregui et al.<sup>16,17</sup> (see Supplementary Methods and Supplementary Fig. 1). Conceptually, this model is an age-stratified compartmental model that describes TB dynamics within a whole, closed population, stratified into 15 age groups during periods of the order of several decades. The model is detailed enough to include demographic evolution and aging, along with heterogeneous contact patterns among age groups that have been adapted from empirical survey studies (see Supplementary Fig. 2).

Here, the model is calibrated to reproduce TB incidence and mortality rates in each country under study for the period 2000–2019, using the burden estimates provided by the WHO. The calibration process gives the diagnosis rate d(t) and the scaled infectiousness  $\beta(t)$ , which are modeled as half-sigmoid-like curves, and, among other parameters, are country-specific. This allows the model to reproduce different epidemiological scenarios. Specifically, the diagnosis rate is defined as:

$$d(t) = \begin{cases} d_0 + (d_{\sup} - d_0)t(t + \frac{1}{d_1})^{-1} & \text{if } d_1 > 0\\ d_0 & \text{if } d_1 = 0\\ d_0 - d_0t(t - \frac{1}{d_1})^{-1} & \text{if } d_1 < 0 \end{cases}$$
(1)

Therefore, the diagnosis rate is parameterized by two quantities  $(d_0, d_1)$ , where  $d_0$  is the value at the beginning of the calibrating window (i.e., year 2000 in this study), and  $d_1$  defines its evolution, either increasing or decreasing with time depending on  $d_1$ 's sign. In the case of a decreasing evolution, the diagnosis rate is bounded to be greater than zero, while in the case of increasing evolution the upper bound is  $d_{sup} = 12.17(\text{year}^{-1})^{16}$ . This latter upper bound corresponds to a minimum diagnosis period of 1 month, assuming that, with a conservative lower boundary, the main symptom of TB is a continuous cough lasting for 3 weeks, followed by a time to diagnose estimated to last at least 10 days<sup>18</sup>. For further details regarding the specific values of epidemiological parameters, calibration processes, and uncertainty estimates, the reader is referred to the original source<sup>16</sup> and the Supplementary Methods.

Once the model is calibrated, we use it to produce forecasts until 2035 under two different scenarios: the baseline scenario, namely, a scenario in which there is no COVID-19 pandemic and thus, no disruption in healthcare systems is introduced, and another one in which a disruption is introduced at the start of 2020 up to the end of 2021, which is the pandemic scenario. During the duration of the pandemic, the diagnosis rate drops according to the reduction observed in the notifications of TB cases that are reported by WHO online, in the last global TB report, and also by the Nikshay program in India. Therefore, the drops in diagnosis rate are country-specific. These drops in TB notifications are fitted to a bump-like asymmetric function, as described through  $d_{red}(t)$  in Eq. (2). This function reproduces the real data and is then applied to the model-calibrated diagnosis rate to produce the diagnosis function under the pandemic scenario. The fitting procedure is a Levenberg-Marquardt Nonlinear Least-Squares using minpack.lm R's

Table 1 Fitted parameters for	diagnosis ı	reduction	in selected
countries.			

Bump parame	ters
Country	$\Theta = \{h, t_1, t_2, k_1, k_2\}$
Indonesia	$\Theta = \{0.494, 0.391, 17.31, 6.226, 117.4\}$
Pakistan	$\Theta = \{0.398, 0.257, 18.04, 3095.3, 826.8\}$
Kenya	$\Theta = \{0.248, 0.859, 5.218, 0.905, 46.51\}$
India	$\Theta_1 = \{0.393, 0.272, 1.25, 134.7, 0.645\}$
	$\Theta_2 = \{0.594, 0.139, 1.069, 3.111, 114.12\}$
The fitting procedu package <sup>19</sup> , where E Indonesia, Pakistan two separate bump $h$ , $k_1$ and $k_2$ are din	re is a Levenberg-Marquardt Nonlinear Least-Squares using minpack.Im R's iq. (2) is applied to the WHO data normalized by the 2019 mean. For and Kenya, one bump is enough for reproducing the data, whereas in India is need to be concatenated, and are denoted here as $\Theta_1$ and $\Theta_2$ respectively. tensionless quantities, whereas $f_1$ and $f_2$ have units of year-1.

package<sup>19</sup>, where Eq. (2) is applied to the data normalized by the 2019 mean for each country:

$$d_{\rm red}(t) = \begin{cases} x1 & t \le t_0 \\ 1 - h \cdot \exp\frac{-k_1(t-t_1)^2}{(t_1 - t_0)^2 - (t-t_1)^2} & t_0 < t \le t_1 \\ 1 - h \cdot \exp\frac{-k_2(t-t_1)^2}{(t_2 - t_1)^2 - (t-t_1)^2} & t_1 < t \le t_2 \\ 1 & t > t_2 \end{cases}$$
(2)

The bump-like function described in Eq. (2) serves as a multiplier to the model-calibrated diagnosis rate, thus, being the diagnosis rate under the pandemic scenario  $D(t) = d(t) * d_{red}(t)$  with  $d_{red}(t) \neq 1$  only during the COVID-19 pandemic, and  $d_{red}(t) = 1$  otherwise. In Table 1 we report the fitted values of each parameter involved in the bump-like description of the TB notification drops.

Finally, during the period of recovery, interventions are aimed at compensating for the drop in diagnosis rates during the pandemic years. We modeled this by multiplying the diagnosis by a scale parameter, as already discussed. Once the recovery period is over, we assume that the diagnosis rate goes back to its original value as given by d(t) up to the end of the simulation.

**Modeling decisions about the disruption**. A conceptually deep limitation of this study that needs to be stressed is that we only describe the effects of COVID-19-induced reductions in TB diagnosis rates and treatment adherence as the main drivers of the interaction between both processes. Admittedly, the effects of the COVID-19 crisis on TB dynamics are more complex than what is described here, and will most likely include alterations in transmission dynamics, effects mediated by economic impact, and long-term damages to health care quality standards beyond diagnosis rates; all these being aspects that lie out the scope of our study, mainly because the relevant data needed to describe the effects of them on TB dynamics are yet to be produced.

Although some of the non-pharmaceutical interventions adopted worldwide have proven their efficacy in reducing COVID-19 spreading<sup>20</sup>, their effectiveness highly depends upon general public adherence and proper knowledge about the pandemic risks. Whereas some studies<sup>21–24</sup> show that the knowledge, attitude, and practice toward COVID-19 basic preventive strategies and conducts are in general positive, there is a great variation between communities and, for example, in India, between socioeconomic levels. Specifically, rural populations, as well as individuals with lower education, and unskilled occupations, are associated with lower scores of knowledge, attitude, and practice toward the basic preventive strategies against COVID-19, which would in turn be expected to contribute to halting TB transmission too<sup>23</sup>. This lack of adherence in the lower socioeconomic levels<sup>25</sup> suggests that it might be misleading to assume that the implementation of countermeasures induces a reduction in the TB force of infection. On the other hand, the changes in mobility due to lockdowns and other restrictions indicate that most of the interactions happen in residential areas (e.g., households) while these interventions are in place (see Supplementary Notes 1 and Supplementary Fig. 3). Admittedly, this could be at the root of some recent observations that report that the number of children diagnosed with TB has increased and that non-pharmaceutical public health interventions likely reduced influenza transmission, but have a lesser effect on *M.tb.* transmission during  $2020^{26-28}$ .

To contextualize our findings in broader scenarios where changes in TB transmission—either toward enhanced or reduced spreading—are considered, we show the results of the basic burden outcomes, incidence, and mortality, in each country, for scenarios in which the transmission is either reduced or enhanced. We also considered an alternative scenario in which the treatment availability is higher than the one adopted in the main text, thus, exploring the effect of an overestimate of the disruption over the treatment (see Supplementary Notes 2 and Supplementary Figs. 4 and 5).

First-line treatment reduction. According to previous reports<sup>3,15</sup>, first-line TB treatment completion has dropped effectively as a consequence of the COVID-19 pandemic, with interruptions in the supply of drugs that delay the start of the treatment in those cases in which the remaining medical capabilities have been enough to diagnose the disease. This inconvenience could not only worsen the expected treatment outcome for the patient but also drive secondary infections even in diagnosed patients if they are not able to quarantine until the treatment could be carried out. We modeled this situation in terms of the epidemiological model by including a fraction of under-treatment pulmonary TB individuals  $(T_p)$  in the expression of the force of infection  $(\lambda(t))$ . On the baseline scenario and without disruptions, those  $T_p$  individuals are not able to contribute to  $\lambda(t)$  as we assume that they are under control by the healthcare system, thus, being controlled and either under quarantine or, later on, medicated with TB drugs that greatly reduce their infectiousness. This means that, under normal circumstances, diagnosed individuals are expected not to be a risk for the rest of the population. However, when disruptions in the supply chain appear, a drop is observed in the first-line and second-line treatments completion<sup>15</sup>, and then diagnosed individuals who are not able to either start the treatment or quarantine could become a risk. For this reason, we obtain an estimate of the fraction of  $T_p$  individuals that contribute to  $\lambda(t)$ ,  $T_{inf}$ , from Cilloni et al.<sup>15</sup> as:

$$T_{\rm inf} = (1 - \eta)T_p \tag{3}$$

where  $\eta = 0.788$ . This value attempts to capture this kind of impact in countries like India and Kenya. It is based on expert opinion in the Stop TB Partnership and USAID about the side effects of the COVID-19 pandemic on TB treatment completion. We assume it to be a good proxy for the real value for the other countries included in this study.

**Reporting summary**. Further information on research design is available in the Nature Research Reporting Summary linked to this article.

#### Results

Forecasts of TB incidence and mortality under COVID-19 pressure. To forecast the effect that disruptions in the diagnostic capabilities and the treatment completion have on TB incidence and mortality trends, we selected four different high-burden

countries, three in Asia (India, Indonesia, Pakistan) and one in Africa (Kenya). Then, we calibrated the mathematical model<sup>16</sup> using the current WHO estimates for TB incidence and mortality rates in those countries and produced forecasts in two separate scenarios. The baseline scenario assumes no disruption, whereas the perturbed scenario incorporates the effects of the pandemic on TB diagnosis and treatment adherence. In the different countries analyzed, the duration of the disruptions has been of variable intensity and length, and while some countries experienced an almost complete return to pre-pandemic levels by June 2021 (India, Pakistan), other countries were still registering lower case notification rates by the end of 2021 compared to values before the COVID-19 irruption.

While treatment adherence is assumed to be reduced a 22% from the pre-pandemic values, according to Cilloni et al.<sup>15</sup>, disruption is introduced based upon available data. These were made publicly available by the WHO for Indonesia, Kenya, and Pakistan<sup>29</sup>, and by the Nikshay governmental program for India<sup>30</sup> during the months—or trimesters, for Pakistan—that followed the irruption of COVID-19. To incorporate those data into our model, we use a piece-wise bump function  $d_{red}(t)$  to model a transient continuous drop in the diagnosis rate trend d(t) that was foreseen within our model upon its calibration on pre-pandemic data (see "Methods", Eq. (1)). Proceeding this way, the actual diagnosis rate in the COVID-19 scenario, D(t), can be obtained as the product of the model-calibrated diagnosis rate and the fitted bump function capturing the disruption due to the COVID-19 pandemic, as shown in Eq. (4) and Fig. 1:

$$D(t) = d(t) * d_{\rm red}(t).$$
(4)

Figure 2 shows the estimated TB incidence per million inhabitants per year in the four countries considered, both in the baseline scenario and considering the negative impact of the COVID-19 pandemic. As observed, a transient surge of TB incidence starts in 2020, which is later foreseen to return to values

close to the baseline trend. In the figure, the dotted line represents the baseline scenario, namely, what would have been the projected evolution of TB incidence without the disruptions of the pandemic. The size of the peak reflects the severity of the saturation of the healthcare system in each country which led to drops of different intensity in diagnosis. The results show that the estimated COVID-19 impact on TB incidence trends is larger in the three Asian countries analyzed than in Kenya. This is a direct consequence of the less severe decays in TB case notifications that have been observed in Africa in comparison to other regions<sup>3</sup>, which have been used to inform our mathematical model. These regional differences, in turn, may be due to a combination of factors. First, as stated by Haider et al.<sup>31</sup>, some of those countries adopted early on measures for facing the pandemic, secondly, COVID-19 has had a smaller effect in Africa, which can be due, in part, to a strong under-diagnosis and partially because its younger population.

Important enough, even if COVID-19 disruptions are assumed to happen only during the pandemic years, the long-term effects span for longer times, sometimes up to 5 years since the start of the COVID-19 pandemic. As observed in Fig. 2, in the long term, TB incidence levels stabilize and recover to their baseline values approximately by the year 2030, resulting in a 10 years window of higher burden that makes the incidence go off the way of TB eradication stated in the End TB Strategy. Moreover, in the absence of any further intervention, the peak of TB incidence caused by the disruptions associated with the COVID-19 pandemic will produce not only new TB cases but also an increase in TB-related deaths all across the world. Specifically, by the end of the simulation period in the year 2035, our model predicts an increase in mortality as shown in Fig. 3, where we have represented both the increment percentage and the total number of accumulated additional deaths between 2020 and 2035. Particularly, we forecast an increase in the number of deaths of 1.28%(1.02-1.62, 95% CI) in India, 2.94%(2.50-3.54, 95% CI) in Indonesia, 0.72%(0.65-0.83, 95% CI) in Kenya and



**Fig. 1 Changes in the diagnosis rate before, during, and after the pandemic period.** We present both the original data and the fitted bumps we use for modeling the disruption, along with the diagnosis rates in two scenarios: the baseline, with no disruption (dotted lines), and the pandemic scenario, where the drop in diagnosis happens and is followed by a return of the diagnosis rates to the baseline scenario. Diagnosis multipliers are obtained directly using the WHO data as the TB notifications in that period divided by the mean of TB notifications in the year 2019. The four countries considered are (**a**) India, (**b**) Indonesia, (**c**) Kenya, (**d**) Pakistan.



**Fig. 2 Projected annual TB incidence in four high-burden countries over the period 2020-2035.** The data-driven model is calibrated with WHO incidence data up to 2019<sup>3</sup>. The shaded area represents the 95% CI and the black line is the median of the model outcome for 500 independent runs of the disrupted scenario. The dotted black line is the model forecast for the scenario in which there was no Covid-19 pandemic. Red dots with error bars are the TB burden provided by the WHO<sup>3</sup> used for calibration. Projected incidence values are calculated at the end of the corresponding year on the *x*-axis. The impact of COVID-19 is modeled as a reduction in diagnosis rates and treatment completion for 2 years (2020 and 2021), see Fig. 1 and main text. The four countries considered are (a) Indonesia, (b) Kenya, (c) India, (d) Pakistan, which account for 42.1% of the total number of TB infections worldwide.



Fig. 3 Model predictions of additional TB-related deaths due to COVID-19-related disruptions in healthcare systems. Error bars are the 95% Cl of 500 independent runs of the model. A Percentage of increase of mortality in comparison to the baseline scenario in 2035 for each of the four countries studied. B Cumulative number of excess deaths caused by the pandemic impact during the whole time window simulated (2020-2035) for each country under study as indicated in the *x*-axis.



Fig. 4 Schematic representation of the increase in diagnosis rate in the post-pandemic scenario. Unlike the situation described in Fig. 1, we consider a compensatory period during which the diagnosis rate is boosted up to  $d(t)^*d_{inc}$ , with  $d_{inc} > 0$ , which applies for the whole duration of the intervention.

1.62%(1.21–2.01, 95% CI) in Pakistan. In absolute terms, the total number of excess deaths could be over 350000 individuals in these four countries alone (Fig. 3B).

Finally, as the current situation with COVID-19 and its variants seems to be far from an end, we explore some alternative scenarios in which a secondary disruption, similar to the one that already happened, is introduced in the model. The results of this exploratory analysis are reported in the supplementary materials (see Supplementary Notes 3 and Supplementary Fig. 6).

Proposed intervention for mitigating the pandemic effect. As shown before, the pandemic circumstances will lead to notable increases in TB incidence and mortality. This represents a critical setback concerning the objective of eradicating TB disease within the next few decades, making it hardly achievable without a rapid and effective recovery strategy. More importantly, the disruptions will cause many preventable deaths. It is thus of utmost importance to elucidate whether new policies could be implemented to revert the negative impact of COVID-19 on TB disease. In what follows, we explore the potential of interventions focused on compensating the decay in diagnosis rates observed during the biennium 2020-2021, through a compensatory boost in the next years, as sketched in Fig. 4. While improvements in passive case finding routinely practice are unlikely to unlock sufficient increases in diagnosis rates, these, combined with the implementation of properly designed strategies of active case finding would constitute the paradigmatic type of interventions capable of producing diagnosis improvements comparable to those here explored.

The potential intervention over the diagnosis of TB cases is modeled using a piece-wise function that combines Eq. (4) with an additional piece introduced after the pandemic disruption is over, and for a parameterized duration,  $T_{rec}^{end} - T_{rec}^{st}$ , to be determined. More specifically, we assume that over this new period of time the pre-pandemic diagnosis rate is effectively increased by a factor  $d_{inc} \ge 1$ , (see also Fig. 4). That is:

$$D(t) = \begin{cases} d(t) * d_{\text{red}}(t) & \text{if} \quad t < T_{\text{rec}}^{\text{st}} \\ d(t) * d_{\text{red}}(t) * d_{\text{inc}} & \text{if} \quad T_{\text{rec}}^{\text{st}} \le t < T_{\text{rec}}^{\text{end}} \\ d(t) * d_{\text{red}}(t) & \text{if} \quad t > T_{\text{rec}}^{\text{end}} \end{cases}$$
(5)

The efficacy of such a compensatory period will in principle be proportional to the intensity and the duration of the intervention. In Fig. 5, we show the impact of considering different combinations of diagnosis boost and duration of the boost on the cumulative excess mortality in 2035.

Clearly, the more intense and longer the additional effort is, the larger the number of averted deaths by the end of the simulation period. As it can be seen, an increase in diagnosis rate during a certain amount of time could eventually revert the negative impact of COVID-19 in TB mortality measured in 2035. More specifically, for all four countries, there is a region in the parameter space for which the increase in mortality in 2035 is close to zero (highlighted contour line of the different panels) if the diagnosis rate of new TB cases increases from 10 to 50% of its original value for a period that spans between 1 and 4 years. Importantly, this implies that the extra death toll that is expected from the effect of COVID-19 on TB diagnosis and treatment during the last 2 years could be fully mitigated if ambitious interventions focused on increasing case detection in the next few years are deployed.

Table 2 reports the number of averted deaths in each country in 2035 when the additional effort is applied for 2-4 years and considering increases in the diagnosis rates of 15, 30, and 45%, starting right after the end of 2022. The reported values are obtained by comparing model forecasts for the estimated number of TB-related deaths in the pandemic scenario with the outcome obtained when the recovery strategy is adopted after the end of the COVID-19 disruptions. The model suggests that it is generally better to increase the diagnosis rate for shorter times than to increase the temporal span and have smaller increments of the diagnosis rate. This is because, in the former situation, more deaths are averted in the long term. Nonetheless, the ideal scenario is still the one in which both dimensions are boosted at the same time, as the longer the time the effort is maintained for a given multiplier, the lower the TB-related death toll caused by the pandemic. As noted before, we stress that for an effort ratio of 1.30 and a temporal span of 4 years, the number of averted deaths almost corresponds to the 100% of additional deaths expected due to the COVID-19 disruptions (see Fig. 3), i.e., the pandemic impact on TB burden could be fully mitigated.

#### Discussion

The COVID-19 pandemic is not over yet, and much remains to be clarified about its impact on the—physical and mental—health of the general population. As of May 2022, the coronavirus SARS-CoV-2 has infected more than 530 million individuals, causing the death of more than 6.3 million people worldwide. Although the SARS-CoV-2 and its associated disease COVID-19 were first identified more than 2 years ago, the scientific community has already been able to describe many of the clinical characteristics and pathogenesis of COVID-19, especially during the acute phase<sup>32,33</sup>. However, there are important features that remain less known, such as the long-term consequences of the disease<sup>34–36</sup> and the relation between comorbidities and their risks upon infection by SARS-CoV-2<sup>37,38</sup>. Another important question that is not fully assessed concerns the indirect effects of the pandemic, and the NPIs adopted for its control and mitigation, over other diseases.

In particular, the large number of healthy individuals that were infected in a very short period, producing the so-called epidemic waves, led to the saturation of many healthcare systems, which in turn induced the implementation of very restrictive measures such as lockdowns and curfews in those countries. These compulsory interventions have been argued to be at the root of important reductions in diagnosis rates of other deadly diseases<sup>5,6</sup>. Yet, the long-term consequences are still to be determined. Here, we have focused on TB, since it is one disease for which disruptions in health care could be most dramatic<sup>13,15</sup> given that even without a pandemic scenario, more than 1.5 million lives are lost every year because of the disease.



Time span (y)

Fig. 5 Relative increase of the expected number of deaths when an intervention is introduced in 2022 with respect to the projected impact of the COVID-19 pandemic. We assume that during the recovery period, whose duration is given by the parameter  $T_{rec}$  (the values of the *X*-axis), there is an increase in the diagnosis rate characterized by a factor  $d_{inc} > 1$  (value of the *Y*-axis). Level (white) curves represent combinations of parameters that give the same excess deaths percentage, as indicated by the values over the curves. The four countries under study are (**a**) Indonesia, (**b**) Kenya, (**c**) India, (**d**) Pakistan.

Using a data-driven epidemiological model<sup>16</sup>, we have quantified the negative impact of COVID-19 on TB diagnosis and its longterm consequences. We have also shown that a rapid and intense post-pandemic intervention could eventually mitigate the expected increase in incidence and mortality of TB. Countries enrolled in this work have been selected because of their high-TB burden, contributing an important amount of cases to the annual TB incidence recorded by the WHO global report. Certainly, all four countries together accounted for a 42.1% of the global TB cases in the year 2019. Individually, India comprises 26.5%, Indonesia accounts for 8.47%, Kenya represents 1.40% and Pakistan is responsible for 5.71% of all cases globally. Additionally, for these countries, the reduction in TB case notification due to the COVID-19 pandemic has been well-documented<sup>3,29</sup>, and spans from milder (Kenya) to more severe magnitudes (Indonesia), which make them suitable case studies to estimate the pandemic negative impact and the design of corrective interventions.

Our results show that a drop in diagnosis rates and first-line treatment compliance statistics leads to a pronounced increase in TB incidence in comparison to the baseline scenario. In turn, the growth in TB burden leads to an upsurge in mortality, producing almost 400,000 excess deaths by 2035 in the four countries of the study combined. However, our study also shows that most of these deaths can still be prevented. In particular, our projections show that an increase of available diagnosis capabilities for some time has very positive effects on long-term TB burden since the pandemic effect can be greatly curtailed. More especifically, if the intervention is powerful and maintained for enough time, the entirety of the expected excess deaths can be avoided. It is worth stressing that the intervention proposed here is aimed at increasing the rate of diagnosed individuals, thus bringing them to treatment as soon as possible. This ultimately points toward cutting TB transmission to a point wherein pre-pandemic burden levels are recovered. As we have demonstrated here, one such intervention could be enough for full mitigation of the negative impact of the COVID-19 pandemic on TB incidence and mortality.

However, it is important to acknowledge that the specific interventions needed to achieve enhancements in the diagnosis

Table 2 Cumulative number of a	averted deaths (i	n thousands) in 2	035 with a post-C	COVID-19 intervention	on initiated in 2022 in
each of the countries studied.					

Country	Diagnosis effort	T = 2 ye	ears	T = 3 ye	ears	<b>T</b> = <b>4</b> ye	ears
Indonesia	1.15	36	(31-43)	52	(44-61)	66	(57-78)
	1.30	67	(57-79)	93	(80-110)	118	(102-139)
	1.45	93	(79-109)	128	(110-150)	160	(139-187)
Kenya	1.15	9	(7-13)	13	(10-19)	17	(12-24)
	1.30	18	(13-25)	25	(18-35)	32	(23-45)
	1.45	25	(18-35)	35	(26-50)	45	(32-63)
India	1.15	121	(98-156)	173	(141-224)	223	(182-288)
	1.30	223	(178-294)	315	(254-418)	402	(326-532)
	1.45	309	(246-419)	432	(347-586)	547	(444-735)
Pakistan	1.15	14	(12-16)	20	(17-23)	27	(23-30)
	1.30	25	(22-29)	36	(31-41)	47	(41-53)
	1.45	35	(30-39)	49	(43-56)	64	(56-71)

rates such as the ones explored in this study-between 10 and 50% increase above basal values-should most likely go beyond policies focused on reducing the diagnostic delay of patients seeking care after experiencing TB symptoms under passive case finding scenarios. Instead, active case finding strategies (ACF) constitute a robust family of interventions that can lead to reductions of both patient<sup>39,40</sup>, and health care system components<sup>41,42</sup> of total TB diagnosis delays that are often compatible with case detection rate improvements similar to those explored here. Along these lines, several epidemiological studies published in the last decade report positive ACF experiences in diverse high-TB burden settings in both rural and urban areas in Africa and Asia alike. For example, in 2020, Vo et al. reported an increase of 15.9% of TB notifications (all-forms) in six districts of Ho-Chi Minh, Vietnam, with respect to another six control districts in the same city<sup>43</sup>. Other studies conducted previously in broad administrative districts in northern Uganda in 2018<sup>44</sup>, and in Cambodia in 2016<sup>45</sup> report results from even more successful ACF strategies, able to increase all-forms TB diagnosis rates up to 30.4% and 46%, respectively, in comparison to control districts. Similar examples also include ACF strategies targeting rural, and even nomad populations, in African countries such as Ethiopia (ref. 46, 98.4% increase in all-forms of TB notification rates in 2013) or Nigeria (ref. 47, 24.5% of increase for all TB notifications among nomadic populations in Adamawa state, 2015). Finally, other studies have made use of mathematical modeling to stress that boosting diagnosis rates through ACF is not only feasible but also cost-effective in the mid to long term<sup>48,49</sup>. These results, taken together, suggest that the implementation of ambitious nationwide strategies of ACF in countries such as the ones studied here could contribute significantly to reducing the TB burden to the extent of mitigating the detrimental effects that COVID-19 has had on the TB epidemics worldwide.

In closing, we also mention that our approach is not exempt from limitations that affect TB transmission models. For instance, the outcome of our model depends on a series of epidemiological parameters and initial burden estimates that are subject to strong sources of uncertainty, thus propagating this uncertainty to the results. This means that future improvements in measuring the input data are expected to impact the quantitative outcomes of our mathematical model, in the same way as it would affect any other model that leans on them. Moreover, in our work, we have only described the disruption caused by the COVID-19 pandemic on the TB care system via a reduction of diagnostic capabilities and treatment availability. Even if these are arguably the primary, and the first effects of the COVID-19 pandemic on TB transmission dynamics that have been characterized, there may be many other effects that are yet hard to parameterize, such as the effect on the transmission that non-pharmaceutical interventions had in the countries that carried them out.

On the one hand, it should be possible, in the near future, to produce more detailed estimates on the disruptions of the pandemic on the complete TB cascade of care, based on the corresponding empiric data disclosed at a greater level of detail than the inputs used in this study. In fact, at the moment of writing this study, there is great heterogeneity in the available empirical data about the effects in the TB cascade of care, which points toward the urge of improving data availability for properly understanding the vast effects of COVID-19 on TB care<sup>50</sup>. Moreover, it is clear that other models, which provide a fine-grain structure capable of reproducing the full TB cascade of care, will be needed for taking advantage of this kind of data, which is certainly unfeasible with the model used in this study. Furthermore, it is well known that the emerging pandemic has disrupted profoundly the age structure of social contacts in human populations worldwide through a combination of mobility restrictions, lockdowns, social distancing, and adaptive conducts driven by self-perceived risk, often associated with the stark variations in susceptibility to severe disease and death that have been extensively reported for COVID-19. All these effects combined have arguably re-wired age-dependent contact structures in a way that is not fully understood, and may not be completely transient. On the other hand, geopolitical and economic shifts driven by the pandemic will for sure exert differential effects on TB transmission dynamics between countries. While the TB modeling community should commit to characterizing these phenomena in depth and incorporate them into model forecasts, these are all questions that remain beyond the scope of this study. Be it as it may, the model projections reported here point toward a worrying scenario about the effects of the current pandemic on TB burden evolution in the near future, regardless of the detailed implementation of the disruptions. As more data on the possibly disparate effects of the COVID-19 pandemic on TB is reported<sup>3,29</sup>, updated modeling scenarios can be considered. Similarly, while the duration of the pandemic has been selected to be the length of the fitted bumps (which is directly related to available data) for all countries under study, the longer this data is

reported, the better quantitative outcomes can be forecasted. Similarly, precise measures of how the pandemic affects the model parameters, such as updated mortality risks or transmission rates, would also increase the quality of the forecasts.

In summary, our work shows that implementing a strategy aimed at boosting TB diagnosis rates after the pandemic holds the promise of mitigating, if not fully reverting, the negative impact of the COVID-19 pandemic on TB excess incidence and mortality, even if that period of boosted diagnosis is transient. While the importance of early diagnosis to arrest TB transmission is well known in TB epidemiology<sup>51,52</sup>, we describe here how pushing that aspect of global TB management strategies in the early postcovid time has the potential of reverting a large fraction of the negative impact caused by the pandemic on the global TB epidemics. Interventions such as chronic cough screenings among people seeking healthcare, or even active screening of TB cases among non-symptomatic individuals, along with protocols targeting specifically pre-clinical and/or smear-negative TB cases do all hold the potential of boosting early diagnosis rates in a way that may well be compatible with the scenarios modeled in this study<sup>51–54</sup>. To prevent the COVID-19 pandemic from destroying all the progress achieved during the last years in global TB control, it is time to prioritize such interventions.

#### Data availability

The data concerning drops in TB notifications and treatment availability are publicly available at the original sources<sup>15,29,30</sup>. The data underlying Figs. 1–5 is present in Supplementary Data 1–5. The data underlying Supplementary Figs. 3–6 is present in Supplementary Data 6–8.

#### Code availability

The code that supports the findings of this study is available from the Zenodo repository<sup>17</sup> and at the following address https://doi.org/10.5281/zenodo.6638450.

Received: 30 November 2021; Accepted: 16 June 2022; Published online: 29 June 2022

#### References

- Borgdorff, M. W., Floyd, K. & Broekmans, J. F. Interventions to reduce tuberculosis mortality and transmission in low-and middle-income countries. *Bull. World Health Organ.* 80, 217–227 (2002).
- Matteelli, A. et al. Tuberculosis elimination: where are we now? *Eur. Respir. Rev.* 27 https://err.ersjournals.com/content/27/148/180035 (2018).
- 3. World Health Organization. *Global Tuberculosis Report 2021* (World Health Organization, 2021).
- Lienhardt, C. et al. Global tuberculosis control: lessons learnt and future prospects. *Nat. Rev. Microbiol.* 10, 407–416 (2012).
- Richards, M., Anderson, M., Carter, P., Ebert, B. L. & Mossialos, E. The impact of the COVID-19 pandemic on cancer care. *Nat. Cancer* 1, 565–567 (2020).
- Ansumana, R., Sankoh, O. & Zumla, A. Effects of disruption from COVID-19 on antimalarial strategies. *Nat. Med.* 26, 1334–1336 (2020).
- Gold, J. A. Covid-19: adverse mental health outcomes for healthcare workers (2020).
- 8. Chen, Y. et al. High SARS-COV-2 antibody prevalence among healthcare workers exposed to COVID-19 patients. *J. Infect.* **81**, 420–426 (2020).
- Fusco, F. et al. Covid-19 among healthcare workers in a specialist infectious diseases setting in Naples, Southern Italy: results of a cross-sectional surveillance study. J. Hosp. Infect. 105, 596–600 (2020).
- Unidad de Investigación en Tuberculosis de Barcelona. The COVID-19 pandemic remains getting worse the tuberculosis control. https://www.uitb. cat/wp-content/uploads/2021/11/nota-prensa-XXV-jornadas-TB\_Eng.pdf (2021).
- McQuaid, C. F. et al. The potential impact of COVID-19-related disruption on tuberculosis burden. *Eur. Respir. J.* 56, 2001718 (2020).
- 12. W. R. O. for Europe. Rapid communication on the role of the GeneXpertt<sup>®</sup> platform for rapid molecular testing for SARS-COV-2 in the WHO European Region (2020).

- 13. Adepoju, P. Tuberculosis and HIV responses threatened by COVID-19. Lancet HIV 7, e319-e320 (2020).
- Malik, A. A. et al. Tuberculosis control and care in the era of COVID-19. *Health Policy Plan.* 35, 1130–1132 (2020).
- Cilloni, L. et al. The potential impact of the COVID-19 pandemic on the tuberculosis epidemic a modelling analysis. *EClinicalMedicine* 28, 100603 (2020).
- 16. Arregui, S. et al. Data-driven model for the assessment of mycobacterium tuberculosis transmission in evolving demographic structures. *Proc. Natl Acad. Sci. USA* **115**, E3238–E3245 (2018).
- Tovar, M. A data-driven TB spreading model that includes coupling with COVID-19. https://doi.org/10.5281/zenodo.6638450 (2022).
- Millen, S. J., Uys, P. W., Hargrove, J., Van Helden, P. D. & Williams, B. G. The effect of diagnostic delays on the drop-out rate and the total delay to diagnosis of tuberculosis. *PLoS ONE* 3, e1933 (2008).
- Elzhov, T. V., Mullen, K. M., Spiess, A.-N. & Bolker, B. minpack.lm: R Interface to the Levenberg-Marquardt Nonlinear Least-Squares Algorithm Found in MINPACK. R package version 1.2. (2016).
- Bo, Y. et al. Effectiveness of non-pharmaceutical interventions on COVID-19 transmission in 190 countries from 23 January to 13 April 2020. Int. J. Infect. Dis. 102, 247–253 (2021).
- Narayana, G. et al. Knowledge, perception, and practices towards COVID-19 pandemic among general public of India: a cross-sectional online survey. *Curr. Med. Res. Pract.* 10, 153–159 (2020).
- 22. Ali, M. et al. Knowledge, attitude, practice and fear of COVID-19: a crosscultural study. medRxiv (2020).
- 23. Tomar, B. S. et al. Indian community's knowledge, attitude, and practice toward COVID-19. *Indian J. Soc. Psychiatry* **37**, 48 (2021).
- 24. Paul, A. et al. Knowledge, attitudes, and practices toward the novel coronavirus among Bangladeshis: implications for mitigation measures. *PLoS ONE* **15**, e0238492 (2020).
- Bhargava, A. & Shewade, H. D. The potential impact of the COVID-19 response related lockdown on TB incidence and mortality in India. *Indian J. Tuberc.* 67, S139–S146 (2020).
- Martinez, L. et al. Transmission of mycobacterium tuberculosis in households and the community: a systematic review and meta-analysis. *Am. J. Epidemiol.* 185, 1327–1339 (2017).
- Aznar, M. et al. Impact of the COVID-19 pandemic on tuberculosis management in Spain. *Int. J. Infect. Dis.* 108, 300–305 (2021).
- Geng, Y., Li, G. & Zhang, L. The impact of COVID-19 interventions on influenza and mycobacterium tuberculosis infection. *Front. Public Health* 9, 672568 (2021).
- WHO. Provisional number of people with new or relapse episodes of TB notified per month. https://worldhealthorg.shinyapps.io/tb\_pronto/ (2022).
- 30. Government of India. Nikshay online tool for monitoring TB control programme (2021).
- Haider, N. et al. Lockdown measures in response to COVID-19 in nine Sub-Saharan African countries. *BMJ Global Health* 5, e003319 (2020).
- Wiersinga, W. J., Rhodes, A., Cheng, A. C., Peacock, S. J. & Prescott, H. C. Pathophysiology, transmission, diagnosis, and treatment of coronavirus disease 2019 (COVID-19): a review. *JAMA* 324, 782–793 (2020).
- Cevik, M., Kuppalli, K., Kindrachuk, J. & Peiris, M. Virology, transmission, and pathogenesis of SARS-COV-2. *BMJ* 371, m3862 (2020).
- Marshall, M. The four most urgent questions about long covid. Nature 594, 168–170 (2021).
- 35. Huang, C. et al. 6-month consequences of COVID-19 in patients discharged from hospital: a cohort study. *Lancet* **397**, 220–232 (2021).
- 36. Sudre, C. H. et al. Attributes and predictors of long covid. *Nat. Med.* 27, 626–631 (2021).
- Sanyaolu, A. et al. Comorbidity and its impact on patients with COVID-19. SN Compr. Clin. Med. 2, 1069–1076 (2020).
- Wang, B., Li, R., Lu, Z. & Huang, Y. Does comorbidity increase the risk of patients with COVID-19: evidence from meta-analysis. *Aging* 12, 6049 (2020).
- Chen, J.-O. et al. Role of community-based active case finding in screening tuberculosis in Yunnan province of China. *Infect. Dis. Poverty* 8, 1–12 (2019).
- Kuznetsov, V. N., Grjibovski, A. M., Mariandyshev, A. O., Johansson, E. & Bjune, G. A. A comparison between passive and active case finding in TB control in the Arkhangelsk region. *Int. J. Circumpolar Health* **73**, 23515 (2014).
- Lorent, N. et al. Community-based active tuberculosis case finding in poor urban settlements of Phnom Penh, Cambodia: a feasible and effective strategy. *PLoS ONE* 9, e92754 (2014).
- Shewade, H. D. et al. Patient characteristics, health seeking and delays among new sputum smear positive TB patients identified through active case finding when compared to passive case finding in India. *PLoS ONE* 14, e0213345 (2019).

## ARTICLE

- Vo, L. N. Q. et al. A comparative impact evaluation of two human resource models for community-based active tuberculosis case finding in Ho Chi Minh City, Vietnam. *BMC Public Health* 20, 1–12 (2020).
- Karamagi, E. et al. Improving TB case notification in Northern Uganda: evidence of a quality improvement-guided active case finding intervention. BMC Health Serv. Res. 18, 1–12 (2018).
- Morishita, F., Eang, M. T., Nishikiori, N. & Yadav, R.-P. Increased case notification through active case finding of tuberculosis among household and neighbourhood contacts in Cambodia. *PLoS ONE* 11, e0150405 (2016).
- Yassin, M. A. et al. Innovative community-based approaches doubled tuberculosis case notification and improve treatment outcome in Southern Ethiopia. *PLoS ONE* 8, e63174 (2013).
- 47. John, S. et al. Tuberculosis among nomads in Adamawa, Nigeria: outcomes from two years of active case finding. *Int. J. Tuberc. Lung Dis.* **19**, 463–468 (2015).
- Mupere, E. et al. Effectiveness of active case-finding strategies in tuberculosis control in Kampala, Uganda. Int. J. Tuberc. Lung Dis. 17, 207–213 (2013).
- Azman, A. S., Golub, J. E. & Dowdy, D. W. How much is tuberculosis screening worth? Estimating the value of active case finding for tuberculosis in South Africa, China, and India. *BMC Med.* 12, 1–9 (2014).
- McQuaid, C. F. et al. The impact of COVID-19 on TB: a review of the data. Int. J. Tuberc. Lung Dis. 25, 436-446 (2021).
- Small, P. M. & Pai, M. Tuberculosis diagnosis—time for a game change. N. Engl J. Med. 363, 1070–1071 (2010).
- Lönnroth, K. et al. Tuberculosis control and elimination 2010-50: cure, care, and social development. *Lancet* 375, 1814–1829 (2010).
- Azman, A. S., Golub, J. E. & Dowdy, D. W. How much is tuberculosis screening worth? Estimating the value of active case finding for tuberculosis in South Africa, China, and India. *BMC Med.* 12, 216 (2014).
- Sekandi, J. N., Neuhauser, D., Smyth, K. & Whalen, C. C. Active case finding of undetected tuberculosis among chronic coughers in a slum setting in Kampala, Uganda. *Int. J. Tuberc. Lung Dis.* 13, 508–513 (2009).

#### Acknowledgements

M.T. acknowledges the support of the Government of Aragón through a Government of Aragón Ph.D. contract order IIU/796/2019. M.T., A.A. and Y.M. acknowledge partial support from the Government of Aragon, Spain and "ERDF A way of making Europe" through grant E36-20R (FENOL), and by Ministerio de Ciencia e Innovación, Agencia Española de Investigación (MCIN/AEI/10.13039/501100011033) Grant No. PID2020-115800GB-100. A.A. and Y.M. also acknowledge support from Soremartec S.A. and Soremartec Italia, Ferrero Group. J.S. acknowledges support from grants PID2019-106859GA-100 and RYC-2017-23560 funded by MCIN/AEI/10.13039/501100011033 and by "ESF Investing in your future". The funders had no role in the study design, data collection, and analysis, decision to publish, or preparation of the manuscript.

#### **Author contributions**

M.T., A.A., J.S. and Y.M. designed research; M.T. performed research; M.T., A.A., and Y.M. analyzed data; M.T., A.A., J.S. and Y.M. discussed results; M.T. wrote the first version of the paper with input from A.A. and Y.M.; and J.S. and Y.M. produced the final version of the manuscript. All authors read and approved the final version of the manuscript.

#### **Competing interests**

The authors declare no competing interests.

#### Ethical approval

The Declaration of Helsinki refers to "research on identifiable human material and data". This study does not use any individual human data. The public, aggregated statistics used in the study cannot be de-identified. Therefore, no ethics approval was needed.

#### **Additional information**

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00145-0.

Correspondence and requests for materials should be addressed to Mario Tovar.

**Peer review information** *Communications Medicine* thanks Yonbing Wang and the other, anonymous, reviewer(s) for their contribution to the peer review of this work. Peer reviewer reports are available.

Reprints and permission information is available at http://www.nature.com/reprints

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2022